MEG Management Committee meeting

30.10.2009, 14:00hrs, West Wing Seminar Room, CBU

Minutes
· Present:  Friedemann Pulvermuller, Rik Henson, Danny Mitchell, Yury Shtyrov, Olaf Hauk, Maarten Van Casteren, Clare Cook, Georgie Morrill.
1. Apologies for absences
· Marie Smith

2. Matters arising 
RA informed the committee that Maxfilter 2.1 should be up and running next week sometime. It has since been installed and YS did some testing.
GM and CC are still looking into the MEG lab outputs which GM is putting on the wiki.

Action: 
RA to install the new version of Maxfilter
GM and CC to look into MEG outputs some more, and GM to get entire list up on the wiki.
3. Minutes of the last meeting
Approved.
4. Computing
FP once again pointed out the need for having Maxfilter 2.1 available, mentioning also its advantages advertised by EN officials at the Isreal Human Brain Mapping Conference in September. RA promised to get Maxfilter installed by Friday, 6th November.
Action: (RA to install the new version of Maxfilter.)
5. Lab business and progress
5.1.  Phantom/empty room recording data: OH reported that this is ongoing.  RH asked if the operators are carrying out the tests each morning and analysing this data. It would be easy to log in on a VNC session to do this.  FP suggested a training session where OH shows the MEG operators, Maarten and Clare, how to use the analysis tool. It was agreed that this will be done and  that in future the empty room recordings each morning will be complemented by running the analysis tool. 
5.2. MvC will liaise with OH and GC to take further the previously started screening for artifacts, especially the 27 Hz artefact possibly due to airconditioning .
Action:
OH to train MEG opertors as described

MvC to arrange for further artefact screening

5.3.  MEG lab output: (Already discussed).
5.4. Other issues: YS reported that the construction of the wall for sound insulation will be commenced in the first week of January 2010, finishing up with the annual service starting on 1st February 2010.  The lab will need to be cleared ready for the work to commence and the machine will need to be warmed up so it suggested to close the lab from 15th December, with flexibility in case there is a particularly high demand to do recordings.
Action: GM to inform UMC/IMC and imagers of lab closure.
RH has noticed the value 4 trigger appearing when it hasn’t been programmed to and wondered if it could be due to the cables, when not plugged into the BNC box, touching together to cause a short circuit for that trigger channel.  YS suggested maybe ordering plastic caps for them to stop this or when doing the empty room recordings, to make sure the trigger channel is on. FP also suggested that everyone should make sure these wires are well connected before every recording and to re-assess this issue at the next meeting.
Action: 
Operators to check trigger channel each moring, when doing empty room recordings.

6. MEG UK meeting at Cardiff Uni

RH has sent a message to all asking if some researchers could attend and present their studies.  FP suggested that who ever would like to go should email the orgnisers at the CUBRIC at Univ Cardiff. FP, RH, YS and OH are thinking of attending.
7. MEG Contingency Plan
CC was sent a copy of this plan from Marion and asked if it was up to date.  YS informed the committee that he was sent it originally.  It contains emergency contact details and details of where information regarding the lab is kept off-site, how much the equipment is worth and how long it would take to replace it.  The committee went through this plan and made a few amendments.  This will then be sent back to Marion.
Action:  CC to amend and return to Marion.

8. MEG lab stats for October
In October, the MEG lab was operational 82 hours, with 168 hours available for experimental studies amounting to 49 percent used time. It looks like November will have an increased usage.
9. AOB
None 
10. Date of the next meeting
Wednesday 25th November 2009
